

Nitrogen use efficiency and summer nitrous oxide emissions in a long term organic grass-clover ley system

Implications of anaerobic digestion of cattle slurry on nitrogen management

Kari Løe

Master's degree project • 30 credits Swedish University of Agricultural Sciences, SLU Department of Soil and Environment Agriculture programme – soil and plant sciences Part number 2025:04 Uppsala 2025

Nitrogen use efficiency and summer nitrous oxide emssions in a long term organic grass-clover ley system - Implications of anaerobic digestion of cattle slurry on nitrogen management

Nitrogeneffektivitet og sommerlystgassutslipp fra et langsiktig økologisk gress-kløver system - Implikasjoner av anaerob utråtning av storfegjødsel på nitrogenforvaltning

Kari Løe

Supervisor:	Sofia Delin, Swedish university of agricultural sciences, department of soil and environment				
Assistant supervisor:	Tatiana Rittl, NORSØK				
Examiner:	Katharina Meurer, Swedish university of agricultural sciences, department of soil and environment				
Credits:	30 credits				
Level:	A2E				
Course title:	Självständigt arbete i Biologi, A2E - Agronomprogrammet - mark/växt				
Course code:	EX0898				
Programme/education: Agriculture programme – soil and plant sciences					
Course coordinating dept:	Department of Soil and Environment				
Place of publication:	Uppsala				
Year of publication:	2025				
Copyright:	All featured images are used with permission from the copyright owner.				
Part number:	2025:04				
Keywords:	Nitrous oxide emissions, Anaerobic digestion, Organic fertilizers, Organic agriculture, Greenhouse gasses, Nitrogen use efficiency				
	Climate change				

Swedish University of Agricultural Sciences

Faculty of Natural Resources and Agricultural Sciences Department of soil and environment

Abstract

This study investigated the long-term impact of anaerobically digested slurry (ADSH) on nitrogen use efficiency (NUE) and nitrous oxide (N₂O) emissions in comparison to untreated slurry (USH) and a control without fertilizer. The hypothesis was that ADSH would improve NUE and reduce N₂O emissions compared to USH but increase emissions relative to the control. The study was conducted on an organic dairy farm in Tingvoll, Norway, where three treatments (control, USH, and ADSH) were applied to experimental plots in a randomized block design. The main outcomes were forage yield, NUE, and N2O emissions, with yield and NUE assessed over multiple years and N₂O emissions measured on residual effects from long-term application during the 2024 growing season when no fertilizer was applied. The results indicated that both USH and ADSH significantly improved yield compared to the control, with ADSH showing a slight yield increase over USH, although this difference was not significant. Yield variations were more influenced by seasonal factors, such as precipitation and temperature, than by fertilizer treatment. NUE was similar between ADSH and USH, and both treatments showed reduced NUE in the second cut compared to the first. A negative correlation between NUE and clover percentage suggested that biological nitrogen fixation may have reduced the efficiency of the applied organic nitrogen fertilizers. N₂O emissions were primarily driven by soil temperature and lagged precipitation, with no significant treatment effect on emissions. Interestingly, ADSH and USH exhibited similar emissions, with the treatment effect only interacting with temperature. The results suggest that environmental factors, particularly temperature and precipitation, have a stronger influence on N₂O emissions than fertilizer type. Limitations of the study include the short duration of N₂O measurements and potential disturbances from frame placement. Overall, the findings suggest that fertilizer type has a minor impact on NUE and N₂O emissions under cool, wet conditions, highlighting the importance of environmental factors in shaping agricultural emissions. Further research is needed to determine if these results hold in other climates or under different management practices.

Keywords: Nitrous oxide emissions, Anaerobic digestion, Organic fertilizers, Organic farming, Greenhouse gases

Table of contents

List of tables	5
List of figures	6
List of equations	8
Abbreviations	9
 Introduction	10 11 12 12 12 13 13 15 15
 Materials and methods Experimental site Experimental design Yield and Nitrogen use efficiency N₂O Flux Measurements Statistics Statistics 2.5.1 Yield and NUE Analyses 2.5.2 Gas Flux Calculations 	17 17 18 18 19 21 22 22
3. Results 3.1 Yield 3.2 NUE 3.3 N ₂ O fluxes	25 25 27 30
4. Discussion 4.1 Yield 4.2 NUE 4.3 N ₂ O-emissions 4.4 Limitations of this study. 4.5 Conclusion	35 36 37 38 38
References	40
Popular science summary	45
Acknowledgements	46

List of tables

Table 1: Total ley yield in kg/ha (combined for cut 1 and cut 2) means and SD for each treatment over 11 years
Table 2: Estimates, standard errors (SE), and p-values (from parametric bootstrapping)for the fixed effects in the total yield model. The effects evaluated include theIntercept (Control treatment), as well as the impacts of Treatment USH andTreatment ADSH compared to the control. The contrast difference betweenADSH and USH is also shown (p-value from pairwise contrast)
Table 3: Random effects for the total yield model. This table presents the variance estimates and standard deviations (SD) for the random effects included in the total yield model. The random effects include Calendar year, Block, and Production year, along with the residual variance
Table 4: Summary of Nitrogen Use Efficiency (NUE) values for each treatment (USH and ADSH) based on two different methods (see materials and methods), NUE1 and NUE2. The table shows the mean and standard deviation (SD) of NUE for each treatment across all years after pooling data from Cut 1 and Cut 2. Both NUE1 and NUE2 are calculated using two differenct equaions.
Table 5: Average clover % per treatment and cut and total along with standard deviations (SD) for 7 years
Table 6: Evaluation of fixed effects in the NUE1 model. The table shows the estimated coefficients, standard errors (SE), t-values, and p-values for the fixed effects in the model. 28
Table 7: Variance estimates for random effects in the NUE1 model. The model includes random intercepts for Plot and Year to account for variation at these levels. The residual variance represents unexplained variation in NUE1
Table 8: Evaluation of fixed effects in the NUE2 model. The table shows the estimated coefficients, standard errors (SE), t-values, and p-values for the fixed effects in the model. 29
Table 9: Variance estimates for random effects in the NUE2 model. The model accountsfor random effects of Plot and Year, with the residual variance capturing theremaining variability in NUE2
Table 10: Mean and standard deviation (SD) of N₂O fluxes (mg/m²/min) for each treatment
Table 11: Cumulative N2O fluxes and standard errors for each treatment during the season of 2024 32
Table 12: N2O model summary of fixed effects 32

List of figures

Figure 1: I	Microbial pathways of N2O-production through nitrification and denitrification. (a) Conversion via hydroxylamine (NH ₂ OH), an intermediate in ammonia oxidation, into N ₂ O. (b) Nitrifier denitrification, where nitrifying bacteria reduce nitrite (NO ₂ ⁻) to N ₂ O under oxygen-limited conditions. (Redrawn from Rapson & Dacres, 2013)1	3
Figure 2: I	Experimental site with climate zones in the North-West of Norway. Colors represent the different climate zones and the red dot issituated at the experimental site "SoilEffects" in Tingvoll.(Picture reworked from World bank, accessed October 2024, https://climateknowledgeportal.worldbank.org/country/norway)	7
Figure 3:	The frames and chambers in some plots in the SoilEffects project. Photo: Kari Løe2	20
Figure 4:N	Manual sampling with syringes and vials for later analysis with gas chromatography. Photo: Johanna Maria Zimmermann	20
Figure 5:0	Gas measurements using GasMet FTIR analyzer. Photo: Johanna Maria Zimmermann2	20
Figure 6:	Total ley yield (combined for both cut 1 and cut 2) by year and treatment. Opaque dots are averages with standard errors and translucent dots are observed yield for each treatment replicate. Missing years (2014 & 2019) there were other crops than ley and no fertilizer was applied2	∍ 25
Figure 7: I	Ley yield for both cut 1 and cut 2 by year and treatment. Opaque dots are averages with standard errors and translucent dots are observed yield for each treatment replicate. Regular lines represent cut 1 and dashed lines represent cut 2. Missing years (2014 & 2019) there were other crops than ley and no fertilizer was applied.	h 26
Figure 8:F	Plot with predicted total (both cut 1 and cut 2) ley yields for 11 years with 95% confidence intervals in color, and beeswarmed (jittered) observed values in grey2	27
Figure 9: /	Average NUE1 and NUE2 per treatment and cut over the years standard error bars. Missing years are due to absence of fertilizing that year or unability to quantify the clover percentage	28
Figure 10:	: Predicted NUE1 and NUE2 for both cuts for USH and ADSH	0
Figure 11:	: N ₂ O-fluxes (solid line) over the season of 2024 with soil temperature for treatments above (dashed line). Ploughing occurred April 30 th and August 6 th .	81
Figure 12:	: Cumulative N2O-emissions over the 2024 season per treatment. CONT - no fertilizer, USH - undigested cattle slurry, ADSH - anaerobically digested cattle slurry. Error bars represents the 95% confidence intervals	32

Figure 13: T	Γhe predicted effect of soil temperature on N2O-fluxes in mg/m2/min b reatment	у 34
Figure 14: T	The predicted effect of lagged precipitation on N2O-emissions in mg/m y treatment.	12/min 34

List of equations

Equation 2 19 Equation 3 19 Equation 4 19 Equation 5 22 Equation 6 23 Equation 7 23	Equation 1	
Equation 3 19 Equation 4 19 Equation 5 22 Equation 6 23 Equation 7 23	Equation 2	19
Equation 4	Equation 3	19
Equation 5	Equation 4	19
Equation 6	Equation 5	22
Equation 7	Equation 6	23
	Equation 7	23

Abbreviations

SLU	Swedish University of Agricultural Sciences
NORSØK	Norges Senter for Økologisk Landbruk / The Norwegian Centre for Organic Agriculture
NUE	Nitrogen use efficiency
N_2O	Nitrous oxide
Ν	Nitrogen
GHG	Greenhouse gas
US	Undigested slurry
ADS	Anaerobically digested slurry
USH	Undigested cattle slurry (treatment code)
ADSH	Anaerobically digested slurry (treatment code)

1. Introduction

In organic agriculture, there are restrictions on agricultural inputs to sustain healthy soils, ecosystems, and human health, while at the same time ensuring reasonable yields. For example, in Norway, the use of mineral fertilizers in organic agriculture is prohibited. Instead, organic farming relies on natural sources of plant nutrients, such as plant residues, nitrogen-fixing plants, animal manure and urine, compost. Additionally, digestate from biogas production may be used, as long as the feedstock used in the biogas process is certified organic. These alternatives help create a more sustainable nutrient cycle in agricultural systems by reducing external mineral nitrogen (N) inputs. Organic N sources are generally considered more environmentally friendly compared to mineral fertilizers due to their lower energy requirements for production and their contribution to nutrient recycling. A meta-analysis of 133 maize studies found that substituting mineral with organic fertilizer increased yield, reduced N losses, and enhanced soil carbon sequestration, resulting in a net carbon sink despite higher CO₂ emissions (Wei et al., 2020). Moreover, the volatility of the mineral fertilizer market, as demonstrated by the mineral fertilizer price crisis in 2022/2023, can heavily impact farmers who rely on these inputs. To reduce dependence on mineral fertilizers, farmers can adopt organic fertilizers like manure, compost or digestate, which recycle nutrients within the farming system, provided the feedstock for the digestate originates from the farm. An additional strategy to decrease reliance on mineral fertilizers is to improve nitrogen use efficiency (NUE) for organic fertilizers, meaning that the fraction of applied N that is actually utlize by the crop is increased. Enhancing NUE requires precise timing of fertilizer application to match crop nutrient demands, which can be challenging when using organic fertilizers, as the nutrients are generally released more unpredictably than from mineral fertilizers. Optimizing NUE reduces N losses to the environment through gaseous emissions and nitrate leaching, the latter of which contributes to eutrophication of water bodies. The introduction of N, whether from organic or mineral sources, alters the N cycle and increases the risk of N losses. However, N is essential for crop production. Therefore, optimizing NUE is essential to minimize N loss, mitigate environmental damage, and improve economic returns for farmers.

1.1 Slurry management in organic milk production

Animal manure is a valuable plant nutrient resource in organic agriculture, making it beneficial for organic farms to integrate animal husbandry with plant production. In Norwegian organicfarming systems, livestock and crop production are typically integrated on the same farm, meaning that manure from organic dairy operations is applied to on-farm ley pastures rather than being transported to separate crop farms. This localized use of manure optimizes nutrient recycling by creating a local circular nutrient system.

Organic dairy farmers employ various slurry management strategies to optimize nutrient utilization and minimize environmental impact. These practices include composting, anaerobic digestion, and direct field application. Composting involves the aerobic decomposition of manure, resulting in a stable product that enhances soil structure and fertility (Bernal et al. 2017). Anaerobic digestion, on the other hand, breaks down manure in the absence of oxygen, producing biogas for energy and digestate that can be used as fertilizer (Rehman et al. 2019). Direct application of raw slurry (or any kind of fertilizer) to fields is also practiced, though it requires careful timing and management to prevent nutrient runoff and leaching, as well as pathogen exposure.

The choice of manure management strategy significantly influences environmental outcomes. For instance, composting can reduce pathogens and weed seeds (Bernal et al. 2017), but if not managed properly, it may lead to ammonia, methane and N₂O (Cao et al. 2019). Anaerobic digestion has also been shown to mitigate methane emissions from manure storage, thereby reducing the overall greenhouse gas (GHG) footprint of dairy operations, although it also increases the risk of ammonia volatilization (El Mashad et al. 2023). Other strategies, such as proper timing and method of manure application, as well as the use of nitrification inhibitors, can also play key roles in minimizing environmental impacts. Understanding the impacts of different manure management strategies on N dynamics is crucial for evaluating their effectiveness in improving NUE and reducing N₂O emissions.

1.2 Nitrogen use efficiency

N is essential for plant growth, being a crucial component of chlorophyll, which is involved in photosynthesis. Therefore, N is a crucial agronomic input for achieving satisfactory crop yields. However, the efficiency with which crops utilize N can vary, as plants do not absorb all the N applied to agricultural soils. When fertilizer application is not properly synchronized with crop nutrient demand, the risk of N losses increase and occur primarly through leaching and gaseous emmisions (N₂O and ammonia) (Ladha et al., 2005). These losses not only reduce crop yield potential but also contribute to environmental issues such as water contamination, increased GHG emissions, and eutrophication in aquatic ecosystems (Ngatia et al., 2019). Nitrogen Use Efficiency (NUE) quantifies how effectively plants utilize applied N. Improving NUE is essential for promoting both agricultural productivity and environmental sustainability. Strategies to enhance NUE include adopting precision agriculture techniques to apply N more accurately, using slow-release or stabilized fertilizers, incorporating organic amendments, and aligning N application with crop growth stages, depending on the specific conditions of the farm (Dobermann & Cassman, 2004). In particular, the use of anaerobically digested cattle slurry, as opposed to raw manure, has shown potential to enhance NUE due to changes in N availability and reduced N losses (Frick et al., 2023). Additionally, biogas treatment of livestock slurry and biowastes has been found to reduce GHG emissions, improve NUE, and decrease N leaching. However, the risks of increased ammonia emissions and methane leakage in digestate needs to be carefully managed to maximize its environmental benefits (Møller et al., 2022).

1.3 Nitrous Oxide (N₂O) emissions

Nitrous oxide (N₂O) is a potent GHG with a long atmospheric lifespan, driving climate change and ozone depletion. Since the industrial revolution, its atmospheric concentration has increased substantially, primarily due to anthropogenic activities, particularly agriculture. With a Global Warming Potential (GWP) 273 times that of CO₂ over 20- and 100-year time frames, and 130 times over 500 years, N₂O plays a major role in global warming and atmospheric chemistry (Forster et al., 2024).

1.3.1 Denitrification

 N_2O emissions from soils occur primarily through the microbial processes of nitrification and denitrification, with denitrification being the dominant pathway (Qui et al. 2022). Denitrification is a respiratory process in which nitrate (NO_3^-) is reduced to gaseous N (N_2) under anaerobic conditions (fig 1). In this process, NO_3^- serves as an electron acceptor, while organic carbon compounds act as electron donors and N_2O is produced as an intermediate product during incomplete denitrification.



Figure 1: Microbial pathways of N2O-production through nitrification and denitrification. (a) Conversion via hydroxylamine (NH₂OH), an intermediate in ammonia oxidation, into N₂O. (b) Nitrifier denitrification, where nitrifying bacteria reduce nitrite (NO₂⁻) to N₂O under oxygenlimited conditions. (Redrawn from Rapson & Dacres, 2013).

Conditions that favor denitrification and N_2O release include high NO_3^- concentrations, abundant organic carbon, and wet or waterlogged (anaerobic) soils (Philippot et al., 2007). These factors are common in agricultural settings where fertilizers, irrigation, organic carbon inputs and soil management practices can amplify N_2O emissions. Quantifying the contribution of each factor to denitrification and N_2O release is critical for designing targeted mitigation strategies to effectively reduce emissions from agricultural soils.

1.3.2 Agricultural Practices and N2O Emissions

Global anthropogenic N₂O emissions have increased significantly between 1980 and 2020, driven primarily by agriculture. The anthropogenic fraction of global N₂O emissions is estimated at 35%, with agriculture contributing 56% of this total (Tian et al., 2024). Manure management and soil emissions, especially from Nbased fertilizers, are major contributors. Another study suggest that agricultural practices account for 60-80% of anthropogenic N₂O emissions (Yang et al., 2021).

Several factors contribute to N₂O emissions from agricultural soils, including N inputs, soil moisture and oxygen availability, soil organic carbon (SOC) content, soil pH, and temperature (Butterbach-Bahl et al., 2013). N inputs enhance

microbial nitrification and denitrification processes that produce N₂O (Li et al., 2020). Soil moisture content, particularly water-filled pore space (WFPS), plays a key role by creating anaerobic conditions that favor denitrification and aerobic conditions that promote nitrification. When irrigation or precipitation increases WFPS, anaerobic conditions prevail, leading to higher N₂O emissions, especially when soil nitrate levels are elevated

SOC can increase N₂O emissions by providing an energy source for denitrifying bacteria, which enhances microbial activity under anaerobic conditions (Adviento-Borbe et al., 2007). Soil pH also influences microbial processes; higher pH can promote complete denitrification, reducing N₂O production, but it may also increase nitrification and N mineralization, which can offset these reductions. Bleken and Rittl (2022) found that raising soil pH from 4.8 to 5.8 halved N₂O emissions from ploughed leys, likely due to enhanced denitrification efficiency. Liming acidic soils can, therefore, help mitigate emissions.

Temperature is another important factor, as warmer conditions accelerate microbial activity, boosting both nitrification and denitrification (Smith et al., 2018). Temperature also promotes the decomposition of organic matter, releasing more carbon and N into the soil.

Tillage practices influence N₂O emissions by aerating the soil, which increases nitrification and N mineralization, while also raising soil temperature and accelerating decomposition. No-till or low-till practices can reduce N₂O emissions by minimizing soil disturbance, although in compacted or poorly drained soils, these practices may lead to waterlogging and increased denitrification (Six et al., 2006). Therefore, it is crucial for farmers to consider specific field conditions when choosing practices to minimize N₂O emissions.

1.4 Anaerobic digestion as a Sustainable Solution

Anaerobic digestion (AD) is a microbial process that breaks down organic matter such as animal manure, crop residues, or food waste in an oxygen-free environment, producing biogas (methane and carbon dioxide) and a nutrient-rich byproduct known as digestate (Rehman et al., 2019). Digestate contains plant nutrients like N, phosphorus (P), and potassium (K), as well as micronutrients essential for plant growth (García-Lopez et al., 2023). The nutrients in digestate are often more concentrated and bioavailable than in the original substrate due to the breakdown process, which releases nutrients as the organic carbon is converted into gas (Tambone et al., 2010). Additionally, the slightly alkaline nature of digestate implies that a significant portion of N is present as ammonium (NH4⁺), which is readily available for plants. While digestate has a lower organic carbon content than raw manure, which can lower soil organic carbon (SOC) levels in the short term (Barłóg et al., 2020), it still contains some organic carbon, unlike mineral fertilizers that lack it entirely. Carbon in digestate ias also more stable, meaning theat SOC can increase in the long term due to less carbon breakdown. The lower C/N ratio of digestate also limits the carbon available for denitrifying bacteria, potentially reducing N₂O emissions compared to raw manure (Tambone et al., 2010). These factors not only reduce N losses via denitrification (thereby decreasing N₂O emissions) but also enhance soil structure and nutrient retention, which ensures that a greater proportion of the available nitrogen is in sync with crop demand, ultimately boosting plant nitrogen uptake and overall nitrogen use efficiency NUE. Frick et al. (2023) suggest that anaerobic digestion (AD) of cattle slurry enhances NUE compared to raw slurry when applied to crops like ryegrass.

Although most research on NUE has focused on short-term effects, the long-term impacts of manure management practices remain largely unexplored. In organic dairy farming systems in Norway, two questions can be asked: How do anaerobically digested cattle slurry (ADSH) and undigested cattle slurry (USH) affect long-term NUE, and what are the residual effects of years of slurry application on N₂O gas emissions, particularly in years when no slurry is applied? This study addresses these gaps by (1) quantifying the long-term change in NUE from ADSH versus USH, and (2) measuring the residual N₂O fluxes to provide the essential estimates needed for sustainable manure management strategies.

In the context of organic dairy farming in Norway, this study seeks to explore how AD may hold the key to improving both environmental and economic sustainability by enhancing NUE while reducing N₂O emissions. If proven useful, AD application might provide a breakthrough for farmers looking to mitigate their environmental footprint.

1.5 Objective and Hypothesis

1.5.1 Objective

One objective of this study is to quantify both the long-term effects of anaerobically digested cattle slurry (ADS) on nitrogen use efficiency (NUE) over several years and its residual impact on N₂O emissions—compared with untreated slurry (US)—during the 2024 growth season (May–October), when no fertilizer is applied. In addition, the study will provide essential estimates of how anaerobic digestion can enhance nitrogen cycling, boost NUE, and reduce N₂O emissions, thereby establishing it as a sustainable alternative to applying untreated cattle slurry in agricultural soils.

1.5.2 Hypotheses

The hypotheses are that:

1. anaerobically digested cattle slurry will increase the NUE in comparison to undigested cattle slurry due to nitrogen being more readily available for plant uptake.

2. the residual effect of long-term application of anaerobically digested cattle slurry will reduce N_2O -emissions compared to undigested cattle slurry, but increase in comparison to the control.

2. Materials and methods

2.1 Experimental site

The experimental site is located in Tingvoll research farm (62°54'N,8°11'E), Møre og Romsdal in North-Western Norway. The research farm is owned by the Norwegian Centre for Organic Agriculture (NORSØK) and the experiment SoilEffects has been ongoing since 2011. The area has a Subarctic climate/Temperate oceanic climate (World bank, fig 2) according to the Köppen-Geiger climate classification system. The annual average precipitation between 2015-2024 (April to October) in Tingvoll was 861 mm, and in 2024 (April to October) it was 872 mm (NIBIO), meaning 2024 was a fairly representative year.





The experiment was set up in 2011 to study the long-term effects of anaerobically digested (AD) slurry compared to undigested slurry (UD) on soil characteristics

and crop yields. The slurry is collected from cattle from NORSØK's own research farm with organic dairy production. The AD slurry is derived from the same UD slurry from the dairy cattle and digested in the farms own biogas reactor. The chemical composition of the AD and UD can be found in Rittl et al. (2023). The system consists of a perennial grass-clover ley harvested 2 times per year, and is re-established every 5 years with a cereal cover crop and harvested as green fodder. During the re-establishment of the ley, no manure is applied (Rittl et al., 2023). In the year 2024 when N₂O emissions were measured, no manure was applied and oats were grown, meaning this study is looking at the residual effects of long-term application.

2.2 Experimental design

The experiment uses a randomized block design consisting of 5 treatments, but here only the following 3 three treatments were used; anaerobically digested slurry (ADSH), undigested slurry (UDSH), and control without any added fertilizer in four (4) blocks which makes 12 plots in total. The fertilized treatments received 220 kg total N per ha and year in the beginning of the growing season between 2011–2023. However, during the years 2014, 2019, 2024, no manure was applied due to re-establishment of the ley and in 2020 due to a problem with the biogas reactor.

2.3 Yield and Nitrogen use efficiency

This study uses data from Rittl et al. (2024) on crop yield, N content, clover percentage collected from the years 2011, 2012, 2013, 2015, 2021, 2022 and 2023. The missing years are due to the cultivation of crops other than ley or the absence of botanical composition evaluations during those years. The botanical composition was conducted using representative samples of 0.5 kg plant material from each plot which was sorted into grass, clover and weeds.

Because of the clover content in the ley, the NUE was calculated two ways; To assess NUEso it is critical to account not only for the N supplied by fertilizer but also for the additional N provided through clover fixation. Therefore, we employ two complementary NUE equations. The first (NUE1) provides a straightforward measure of NUE by comparing the total crop N harvested from fertilized plots to that from unfertilized plots, relative to the rate of fertilizer N applied. The second (NUE2) refines this measure by correcting for the N added through clover fixation. This dual approach enables us to disentangle the effects of direct fertilizer application from the biological contributions of clover

Equation 1

$$NUE1 = \frac{(NF) - (NC)}{R}$$

Equation 2

$$NUE2 = \frac{(NF) - (NC)}{(R + CF) - (CC)}$$

In these equations, **NF** denotes the total crop N harvested from fertilized plots, while **NC** denotes the total crop N harvested from unfertilized plots. The variable **R** represents the rate of fertilizer N applied to the fertilized plots. In the second equation, **CF** represents the amount of N added via clover fixation in fertilized plots, and **CC** represents the N fixed by clover in unfertilized plots.

For estimating clover N-fixation, equations by Nyborg (1995) were used. The equations are adjusted based on the legume proportion in the crop. The rationale for employing two versions is that the efficiency of N fixation by clover can vary with its relative abundance in the ley. When clover makes up less than 50% of the crop, a slightly higher efficiency factor is used, whereas a lower factor is applied when clover dominates (>50%).

Legume proportion < 50 %:

Equation 3

Nfix = DS crop x % clover x 0,0345 x (0,9/100)

Legume proportion > 50 %:

Equation 4

 $Nfix = DS \, crop \, x \, \% \, clover \, x \, 0,0345 \, x \, (0,85/100)$

Here, DS_crop represents the dry substance of the crop.

2.4 N₂O Flux Measurements

During the growth season of 2024 there was not ley, but oats grown in the experiment. Therefore, no manure was applied this year, and the gas analysis is done on the residual effect of previous slurry and digestate application through several years. The plots were ploughed on April 30th and August 6th. This is not a common occurrence, but due to a high weed pressure, so mechanical weeding had to be done. The plots measured 8 x 3 m and the frames were 51 x 51 cm. One frame was positioned in each plot (Figure 3), and the edges filled with water to avoid gas transfer between outside and inside the chamber. The chamber lids were placed on the frames at the start of gas measurement.



Figure 3: The frames and chambers in some plots in the SoilEffects project. Photo: Kari Løe

Two methods were used for gas flux measurements: the GasMet FTIR analyzer (figure 5) and gas chromatography with manual vial sampling (figure 4). The GasMet and gas chromatographymeasures a variety of gases, but this study only focuses on N₂O. For both methods gas samples were collected using one manual static chamber per plot. The GasMet system was used in April and May 2024, to provide real-time gas concentration data directly in the field. From June to October 2024, gas samples were collected manually using glass vials for later analysis.



Figure 5: Manual sampling with syringes and vials for later analysis with gas chromatography. Photo: Johanna Maria Zimmermann



Figure 4:Gas measurements using GasMet FTIR analyzer. Photo: Johanna Maria Zimmermann

For vial sampling, glass vials were marked with the date and vial number and evacuated using a vacuum machine before sampling. Gas samples were collected using syringes with 2-Way stopcocks. To ensure proper mixing of the chamber air, the syringe was plunged five times before injecting the sample through the septum into the vial. The vials were over-pressurized to maintain shelf stability. Four gas samples were taken per plot at 10-minute intervals (0, 10, 20, and 30 minutes) after chamber closure to allow for gas flux calculations. These vials were sent to Germany for analysis at Rheinland-Pfälzische Technische Universität Kaiserslautern-Landau (Rhineland-Palatinate Technical University Kaiserslautern-Landau) using a gas chromatograph equipped with a pressure sensor.

For all sampling events, weather conditions, soil temperature at 5 cm depth in each plot, and the time of measurement were recorded. Approximately two measurements were taken per week, depending on weather conditions (preferably after rain), resulting in a total of 27 sampling dates during the 2024 season.

2.5 Statistics

All statistical analyses were conducted in R (version 4.4.2; R Core Team, 2024) within the RStudio IDE (version 2024.09.1+394). The following R packages were used:

- tidyverse (Wickham et al., 2019): Data manipulation and visualization.
- lubridate (Grolemund & Wickham, 2011): Handling date-time data.
- performance (Lüdecke et al., 2021): Model diagnostics.
- lme4 (Bates et al., 2015): Mixed-effects models.
- pbkrtest (Halekoh & Højsgaard, 2014): Bootstrapping and Ftests in mixed models.
- readxl (Wickham & Bryan, 2023): Importing Excel data.
- ggbeeswarm (Clarke et al., 2023): Beeswarm plots.
- ggpubr (Kassambara, 2023): Publication-ready ggplot2 figures.
- ImerTest (Kuznetsova et al., 2017): p-values for mixed-effects models.
- emmeans (Lenth, 2024): Post-hoc analysis of estimated marginal means.
- gasfluxes (Fuss et al., 2024): Gas flux calculations.
- gridExtra (Auguie, 2017): Arranging multiple plots.

Significance was set to p < 0.05.

2.5.1 Yield and NUE Analyses

Yield and NUE data from all plots and treatments were combined into a single dataset. Relevant variables were extracted, and the data were processed to ensure consistency. Any missing values were identified and removed using the drop_na() function.

A linear mixed-effects regression model was developed to predict yield, incorporating treatment as a fixed effect and plot, production year (the age of the ley) and calendar year as random effects. Predictions were generated for all possible combinations of Treatment, Year, and Plot. Final model:

```
Total_yield ~ Treatment + (1 | Produc_year_ley) + (1 |
Block) + (1 | Year)
```

Two separate mixed-effects models were constructed to estimate NUE1 and NUE2. The first model for NUE1 included treatment, harvest cut (first or second harvest), and Clover percent as predictors, with plot and calendar year as random effects. The second model for NUE2 used the same predictors. The difference between the two models is the way of calculating NUE using two different equations (see section 2.4). Interaction terms were assessed using ANOVA comparisons, and non-significant terms were removed. I used PBmodcomp to perform a parametric bootstrap likelihood ratio test comparing nested mixed-effects (e.g., interaction terms). This method yields more reliable p-values than standard tests for mixed models. Final models:

NUE1 ~ Treatment + Cut + Clover percent + (1 | Plot) + (1 | Year)

NUE2 ~ Treatment + Cut + Clover_percent + (1 | Plot) + (1 | Year)

2.5.2 Gas Flux Calculations

Gas flux data were analyzed in R. Data from multiple sampling dates were combined into a single dataframe, and missing values were excluded. A one-day lag for precipitation (dplyr::lag()) was included to assess delayed rainfall effects on N₂O fluxes. To calculate gas flux in mg/m²/min, gas concentrations measured in ppm are first converted to mg/m³ using the ideal gas law. Specifically, the conversion is given by:

Equation 5

$$Cmg/m3 = \frac{Cppm \times MW \times P}{R \times T}$$

where:

- Cppm is the gas concentration in ppm,
- MW is the molecular weight of the gas (g/mol),
- P is the ambient pressure (kPa),
- R is the universal gas constant (8.314 kPa \cdot m³·mol⁻¹·K⁻¹), and
- T is the ambient temperature (in Kelvin).

This mass concentration is then used to calculate the flux by accounting for the chamber geometry and time interval. The flux is calculated as:

Equation 6

$$Flux (mg/m^2/min) = \frac{Cmg/m3 \times V}{A \times \Delta t}$$

where:

- V is the chamber volume (m³),
- A is the chamber's cross-sectional area (m²), and
- Δt is the time interval (min) over which the concentration change is measured.

Combining the two steps, the overall equation becomes:

Equation 7

$$Flux (mg/m^2/min) = \frac{\left(\frac{Cppm \times MW \times P}{R \times T}\right) \times V}{A \times \Delta t}$$

Flux calculations were performed using a linear mixed regression model implemented in the gasflux package.

Then, different transformations of the data were tested by fitting different linear models (LMs) to find the form of the data that best meets the model assumptions and provides the most robust and interpretable results. The transformations included natural logarithmic, base-10 logarithmic, base-2 logarithmic, squre root and cube root transformations. After evaluating the model performances by using model diagnostics the moel using the natural logarithmic transformation was selected. Then, interaction models were built using treatment, soil temperature, and lagged precipitation, testing all pairwise interactions. Non-significant interactions were removed to simplify the model.

The final model for N₂O fluxes included treatment, soil temperature, and lagged precipitation as fixed effects, with plot and sampling date as random effects:

N2O_flux_nlog ~ Treatment * soiltemp + lagged precipitation + (1|plot) + (1|date)

Where N2O_flux_nlog is the natural logarithm of the N₂O flux (mg/m²/min), soiltemp is the centered soil temperature, lagged precipitation is the centered lagged precipitation (precipitation one day before the sampling), (1|plot) is the random effect of plot and (1|date) is the random effect of the sampling date.

Predictions were generated for varying soil temperature and precipitation scenarios, and visualizations with 95% confidence intervals were created using ggplot2.

3. Results

3.1 Yield

Mean total yields over the years showed that both ADSH and USH consistently produced higher yields than the Control (table 1, figure 6). However, all in all over the years, the differences between ADSH and USH were non-significant (Figure 6, Figure 7).

Table 1: Total ley yield in kg/ha (combined for cut 1 and cut 2) means and SD for each treatment over 11 years.

Treatment	Mean yield	SD	
Control	5117	1513	
USH	9814	1544	
ADSH	10084	1513	



Figure 6: Total ley yield (combined for both cut 1 and cut 2) by year and treatment. Opaque dots are averages with standard errors and translucent dots are observed yield for each treatment replicate. Missing years (2014 & 2019) there were other crops than ley and no fertilizer was applied.

When splitting up the yields into the two yearly harvests (cut 1 and cut 2) Yield differences appear to be more strongly influenced by cut than by treatment, as seen in Figure 7.



Figure 7: Ley yield for both cut 1 and cut 2 by year and treatment. Opaque dots are averages with standard errors and translucent dots are observed yield for each treatment replicate. Regular lines represent cut 1 and dashed lines represent cut 2. Missing years (2014 & 2019) there were other crops than ley and no fertilizer was applied.

Table 2: Estimates, standard errors (SE), and p-values (from parametric bootstrapping) for the fixed effects in the total yield model. The effects evaluated include the Intercept (Control treatment), as well as the impacts of Treatment USH and Treatment ADSH compared to the control. The contrast difference between ADSH and USH is also shown (p-value from pairwise contrast).

Fixed Effect	Estimate	SE	p-value
Intercept (Control)	5117.16	378.47	< 0.001
Treatment USH	4696.76	244.31	< 0.001
Treatment ADSH	4966.73	244.31	< 0.001
ADSH vs. USH	-270.0	246	0.5174

Table 3: Random effects for the total yield model. This table presents the variance estimates and standard deviations (SD) for the random effects included in the total yield model. The random effects include Calendar year, Block, and Production year, along with the residual variance

Random Effect	Variance Estimate	SD
Calendar year	1 057 000	1027.87
Block	69 390	263.42
Production year	0.158	0.40
Residual	1 313 000	1145.91



Figure 8:Plot with predicted total (both cut 1 and cut 2) ley yields for 11 years with 95% confidence intervals in color, and beeswarmed (jittered) observed values in grey

The model revealed significant differences in yield among treatments. The intercept, representing the baseline treatment (Control), was estimated at 5117 kg/ha/year (p < 0.001). Both USH and ADSH treatments resulted in significantly higher yields relative to the Control, with increases of 4697 kg/ha/year and 4967 kg/ha/year, respectively; however, there was no statistically significant difference between USH and ADSH. These results indicate that the yield differences in the experiment are primarily driven by the treatment effects, while variability associated with production year, block, and year is accounted for as random effects. The model does not include direct measurements of environmental factors such as temperature or soil moisture, so the yield estimates reflect the average conditions captured by the random effects rather than specific scenarios of temperature or moisture extremes.

3.2 NUE

For both treatments (USH and ADSH), the mean values for NUE1 and NUE2 are similar (around 0.4 for both methods). This suggests that both methods yield comparable results for the treatments (table 4). Overall, the data suggests that both treatments (USH and ADSH) have similar NUE values on average, but the degree of variability in NUE1 and NUE2 differs, with ADSH having a bit more variation in both methods compared to USH. The control had most clover percentage followed by ADSH and USH (Table 5)

Table 4: Summary of Nitrogen Use Efficiency (NUE) values for each treatment (USH and ADSH) based on two different methods (see materials and methods), NUE1 and NUE2. The table shows the mean and standard deviation (SD) of NUE for each treatment across all years after pooling data from Cut 1 and Cut 2. Both NUE1 and NUE2 are calculated using two differenct equaions.

Treatment	Mean NUE1	SD1	Mean NUE2	SD2	
USH	0.405	0.141	0.382	0.151	
ADSH	0.409	0.201	0.391	0.197	

Table 5: Average clover % per treatment and cut and total along with standard deviations (SD) for 7 years

Treatment	Cut 1	SD Cut1	Cut 2	SD Cut2	Total	SD Total
Control	32.2	14.4	33.1	16.6	65.3	22.0
USH	23.3	16.3	13.5	11.0	36.9	19.6
ADSH	24.3	18.3	17.4	12.7	41.6	22.3

Mean NUE using two methods (with or without clover N-fixation) is shown for both cuts over the years in figure 9. There was no evidence for differences between the two treatments.





Figure 9: Average NUE1 and NUE2 per treatment and cut over the years standard error bars. Missing years are due to absence of fertilizing that year or unability to quantify the clover percentage

Table 6: Evaluation of fixed effects in the NUE1 model. The table shows the estimated coefficients,
standard errors (SE), t-values, and p-values for the fixed effects in the model.

();	· 1 J	0 00			
Fixed effect	Estimate	SE	t-value	p-value	
Intercept (/Cut1)	0.29019	0.035219	8.240	< 0.001	
treatmentUSH	-0.0044	0.041426	-0.106	0.919	
Cut2	-0.1666	0.017464	-9.538	< 0.001	

-2.873

Table 7: Variance estimates for random effects in the NUE1 model. The model includes random intercepts for Plot and Year to account for variation at these levels. The residual variance represents unexplained variation in NUE1.

Random effect	Variance	SD
Plot	0.002888	0.05374
Year	0.002123	0.04607
Residual	0.007468	0.08642

The NUE1 model showed that the intercept for ADSH and Cut 1 was highly significant (p < 0.001), with a baseline NUE of 0.29019 (table 6). There was no significant difference between ADSH and USH (estimate = -0.0044, p = 0.919). However, Cut 2 had a significantly lower NUE1 compared to Cut 1 (estimate = -0.1666, p < 0.001), indicating a decline in N use efficiency in the second harvest. Additionally, Clover percentage had a significant negative effect on NUE1 (estimate = -0.0317, p = 0.005), suggesting that higher clover presence was associated with lower NUE1 values. The random effects (table 7) showed that variation across plots (σ^2 = 0.0029) was slightly higher than across years (σ^2 = 0.0021), but both were relatively small. The residual variance (σ^2 = 0.0075) remained the largest source of unexplained variation, suggesting that factors beyond those included in the model contribute to differences in NUE1.

Table 8: Evaluation of fixed effects in the NUE2 model. The table shows the estimated coefficients, standard errors (SE), t-values, and p-values for the fixed effects in the model.

Fixed effect	Estimate	SE	t-value	p-value
Intercept (ADSH/Cut1)	0.27886	0.03237	8.615	< 0.001
treatmentUSH	-0.00481	0.03927	-0.122	0.907
Cut2	-0.16041	0.01644	-9.758	< 0.001
Std. Clover percentage	-0.04400	0.010240	-4.297	< 0.001

Table 9: Variance estimates for random effects in the NUE2 model. The model accounts for random effects of Plot and Year, with the residual variance capturing the remaining variability in NUE2.

Random Effect	Variance	SD
Plot	0.002598	0.05097
Year	0.001458	0.03818
Residual	0.006592	0.08119

The NUE2 model showed similar trends, with a significant intercept for ADSH and Cut 1 (0.27886, p < 0.001) (Table 8). Again, there was no significant difference between ADSH and USH (estimate = -0.00481, p = 0.907). As in NUE1, NUE2 was lower in Cut 2 (estimate = -0.16041, p < 0.001), and increasing clover percentage further decreased NUE2 (estimate = -0.04400, p < 0.001). Random effects (table 9) showed that variation across plots ($\sigma^2 = 0.0026$) was

slightly higher than across years ($\sigma^2 = 0.0015$), but both were relatively minor. The residual variance ($\sigma^2 = 0.0066$) remained the dominant source of variation, similar to NUE1, indicating that unexplained variability is still present in the model.

Both models showed consistent results: USH and ADSH had no significant differences in NUE, while Cut 2 consistently had lower NUE than Cut 1, consistent with the lower biomass yield from Cut 1 to Cut 2. Additionally, higher clover percentages negatively impacted NUE in both models (figure 10). While random effects contributed some variation, residual variance remained the largest source of unexplained differences, highlighting the potential influence of additional factors not included in the model.



Figure 10: Predicted NUE1 and NUE2 for both cuts for USH and ADSH

3.3 N₂O fluxes

N₂O fluxes varied across treatments, with ADSH showing the lowest mean emissions, while USH and CONT exhibited higher fluxes (table 10). Variability in emissions was highest in the CONT treatment (table 10).

Table 10: Mean and standard deviation (SD) of N_2O *fluxes (mg/m²/min) for each treatment.*

Treatment	Mean N ₂ O flux (mg/m ² /min)	SD
Control	0.000716	0.00115
USH	0.000734	0.000644
ADSH	0.000526	0.000517

Figure 11 shows the N_2O -fluxes over the season of 2024 with the mean soil temp measured at 5 cm depth for each treatment.



Figure 11: N_2O -fluxes (solid line) over the season of 2024 with soil temperature for treatments above (dashed line). Ploughing occurred April 30th and August 6th.



Figure 12: Cumulative N2O-emissions over the 2024 season per treatment. CONT - no fertilizer, USH - undigested cattle slurry, ADSH - anaerobically digested cattle slurry. Error bars represents the 95% confidence intervals.

Mean cumulative flux was lowest for ADSH (0.117 mg/m^2), followed by USH (0.148 mg/m^2) and CONT (0.162 mg/m^2) (table 11). However, treatment effects were not statistically significant (figure 12, table 12).

Treatment	Cumulative flux (mg/m ² /min)	Cumulative SE
Control	0.162	0.0132
USH	0.148	0.0237
ADSH	0.117	0.00877

Table 11: Cumulative N2O fluxes and standard errors for each treatment during the season of 2024

The model results for N₂O fluxes (Table X) showed that soil temperature had a strong positive effect on N₂O flux (estimate = 0.124, p < 0.001), indicating that higher temperatures led to increased emissions. Lagged precipitation also had a strong positive effect (estimate = 0.026, p = 0.002). However, there was no evidence for main treatment effects.

Table 12: N2O model summary of fixed effects

Term	Estimate	p-value	Significance
Intercept (control)	-7.213	< 0.001	Highly significant

treatmentADSH	-0.053	0.765	Not significant
treatmentUSH	+0.096	0.590	Not significant
cen_soiltemp	0.124	< 0.001	Highly significant
cen_precip_lag	0.026	0.002	Significant

A significant interaction between treatment and soil temperature (p < 0.001) indicated that the relationship between temperature and N₂O flux varied by treatment (figure 13), meaning that the effect of temperature on N₂O emissions depended on the fertilizer treatment. In contrast, the interaction between treatment and lagged precipitation was not significant (p = 0.427), meaning that the effect of precipitation on N₂O flux did not depend on fertilizer treatment (figure 14), thus, this interaction term was not included in the model. Lagged precipitation in this model represents the total precipitation from the previous day, based on the assumption that N₂O flux responses may be delayed rather than immediate. The model showed greater variation in N₂O fluxes across sampling dates ($\sigma^2 =$ 0.069) than between plots ($\sigma^2 = 0.052$), indicating temporal variability had a stronger influence. Residual variance remained high ($\sigma^2 = 0.183$), suggesting other unaccounted factors affected fluxes.



Figure 13: The predicted effect of soil temperature on N2O-fluxes in mg/m2/min by treatment



N2O Flux vs. lagged precipitation by Treatment

Figure 14: The predicted effect of lagged precipitation on N2O-emissions in mg/m2/min by treatment.

4. Discussion

This study aimed to assess the impact of long-term cattle slurry application of anaerobically digested slurry (ADSH) on NUE as well as N₂O emissions in comparison to untreated slurry (USH). Specifically, we hypothesized that ADSH would enhance NUE due to increased N and decreased organic carbon availability in the digestate and that ADSH would reduce N₂O emissions compared to USH but increase emissions relative to the control.

4.1 Yield

The results from the yield analysis show that both USH and ADSH outperform the control, making them both viable alternatives for organic fertilizer. Although ADSH showed a marginal yield increase (4967 kg/ha) compared to USH (4697 kg/ha), this effect could have arisen due to random variation rather than true treatment effect as the difference was not statistically significant. This is consistent with a study that found no significant difference in forage yield between digested and undigested slurry applications over a three-year period (Walsh et al. 2018). In contrast, short-term studies that have reported a pronounced yield increase from digestates attribute this to their higher content of readily available ammonium (Möller & Müller 2012). However, in this long-term study the data suggest that any initial advantage of ADSH diminishes over time, likely due to the mineralization of organically bound N in the USH, resulting in similar yield performances between ADSH and USH long-term. The clover percentage was highest for the control treatment, likely due to the fact that N fixing plants do well without N fertilizer while grasses do not, leading to clover taking over in the control treatment at a higher rate than in the fertilized treatments.

The yield analysis also revealed that the yields differed substantially between cut 1 and cut 2, where yield for cut 1 was larger. This is consistent with previous findings where the first cut is generally larger than the second. A study based in Finland by Niskanen et al. (2006) showed that the yield for the second cut for several Timothy varieties dropped from the first cut. This yield decrease between the first and second harvest is a common occurrence, with lower N availability and shorter growth period. Yield differences were differed more between years than between treatments, suggesting that seasonal factors play a bigger role than

these particular fertilizer treatments. For example, 2018 had a large yield dip, likely due to the massive drought that year. The random effects emphasized the importance of accounting for variability across years and blocks, which accounted for considerable yield variation. Furthermore, the residual variance reained high, suggesting that there are other factors influencing yield that were not included in this model analysis. Future studies should investigate whether fertilizer application rate, annual precipitation, or temperature variations significantly contribute to yield differences.

Based on this study, the choice between ADSH and USH may not significantly impact yield performance, and it is therefore no evidence to warrant a fertilizer choice based on yield performance alone.

4.2 NUE

The results did not support the hypothesis that ADSH would significantly improve NUE compared to USH. Both treatments exhibited similar NUE values, indicating that the presumed greater N availability in ADSH did not translate into increased uptake efficiency under these field conditions. Consistent with the yield analysis, NUE was significantly lower in the second cut compared to the first. This suggests a seasonal effect on N uptake efficiency, potentially due to lower N availability later in the growing season. I was unable to find studies who compared NUE between two cuts in a ley system, so this may be an important observation.

Another interesting observation was the negative correlation between NUE and clover percentage, suggesting that increased biological N fixation from clover may have reduced the relative efficiency of applied N fertilizers. This finding aligns with previous studies showing that biological N fixation can contribute to soil N pools but may not necessarily translate into increased crop NUE. A study by Elgersma and Hassink (1997) found that although legume plants can fix atmospheric N, the transfer to non-legume plants is low. In the calculation it can then seem like NUE is lowered, because the legume fixation is regarded as an input, but doesn't necessarily contribute to the grass N uptake.

The choice of NUE calculation method also might have played a role in interpretation. NUE1 provides a straightforward estimate of fertilizer-derived N use, whereas NUE2 accounts for biological N fixation, making it more applicable in systems with substantial legume contributions. However, in this study, the difference between NUE1 and NUE2 was minimal, suggesting that the method of calculation did not substantially alter conclusions.

4.3 N₂O-emissions

The analysis of N₂O emissions indicated that soil temperature and lagged precipitation (precipitation from the day before) were the primary drivers of N₂O fluxes with warmer and wetter conditions leading to increased emissions. This aligns with existing literature that shows temperature as a stimulating factor for microbial activity (Smith et al., 2018), and wet conditions as a driver of denitrification (Philippot et al., 2007).

Contrary to the hypothesis, the residual effect of long term application of ADSH did not significantly reduce N_2O emissions compared to USH and did not increase compared to the control. This suggests that in cold, wet climates such as Norway, environmental factors, specifically temperature and precipitation, may have a stronger influence on emissions than fertilizer type alone. This could differ in warmer climates where microbial activity may be constantly higher due to higher soil temperature. However, interestingly, the effect of temperature depended on fertilizer treatment. This could be because the control treatment had much more scarce foliage, meaning there was less insulation against incoming sun, leading to higher temperature fluctuations and thus higher emissions.

Overall, N₂O fluxes were primarily driven by soil temperature and lagged precipitation, with treatment effects emerging only in interaction with temperature. The absence of a direct treatment effect suggests that ADSH and USH did not independently alter N₂O fluxes but may influence emissions under specific temperature conditions. The significant random effects underscore the need to account for both temporal and spatial variability in emissions. Despite similar temperature measurements across treatments (Figure 11), the significant interaction (p < 0.001) indicates that the relationship between soil temperature and N₂O flux differs by treatment. This suggests that even subtle variations in the temperature–flux relationship can result in a statistically significant interaction if the regression slopes differ. Additionally, response variability and a large sample size may contribute to the detected significance, even if visual differences appear minimal. Further diagnostics, such as plotting regression lines with confidence intervals, could help clarify these treatment-specific effects.

The use of a one-day lag for precipitation was based on previous research showing that N₂O fluxes often peak shortly after rainfall due to increased soil moisture and microbial activity. However, it is possible that longer lag periods (e.g., two or more days) could also influence N₂O emissions. A sensitivity analysis testing different lag durations would help assess whether delayed precipitation effects play a larger role than currently estimated.

There was also still a large amount of residual variance in the model, suggesting that additional factors, beyond those considered in the analysis, may be influencing emissions. One major possible contributor to the residal variance could be the ploughing on the 30th of April and 6th of August. Perhaps including "days from last ploughing" in the model could eliminate some of the variance.

Potential other unaccounted factors include soil texture variations, differences in microbial community composition, and localized drainage patterns, all of which could influence N₂O emissions. An additional consideration is the potential impact of frame placement on N₂O flux measurements. Due to the presence of large stones, frames had to be repositioned multiple times before installation, leaving behind small soil disturbances. These disturbances may have influenced N₂O emissions by altering soil aeration and microbial activity in localized areas, potentially contributing to the observed residual variance in emissions.

4.4 Limitations of this study

The N₂O emissions were only measured during a single season (April-October 2024). Measuring over several years could potentially give a more reliable result and perhaps eliminate some residual variance. It may also be wise to stick with one measurement method for gas sampling, as two different methods were used in this study, which could impact the results of the gas fluxes. It would also be interesting to measure N₂O emissions in a year with fertilizer applied to determine whether treatment differences become more pronounced under active fertilization. Since previous literature highlights fertilizer application is a primary driver of N₂O emissions, the absence of fertilizer this year may have masked potential treatment effects even though there has been long-term previous application.

The relationship between NUE and N₂O emissions in organic farming systems is complex and influenced by multiple factors. In general, improving NUE by enhancing N uptake can reduce the amount of N left in the soil, which could in theory lower the potential for N₂O emissions. However, in this study, the role of environmental factors, such as soil temperature and moisture, appears to be more important in driving N₂O emissions. In line with studies by Smith et al. (2018), our results showed that soil temperature and precipitation were the primary drivers of N₂O emissions, with temperature acting as a stimulant for microbial activity and moisture promoting denitrification processes by increasing anaerobic volume in the soil. Therefore, even if NUE improves, N₂O emissions may still be driven by environmental factors, which warrants further research into how we can mitigate these emissions in organic farming.

4.5 Conclusion

This study assessed the impact of anaerobically digested slurry (ADSH) on yield, NUE and N₂O emissions, comparing it to untreated slurry (USH) and a control. Yield and NUE were measured over several years, while N₂O emissions were measured this year, focusing on residual effects as no fertilizer was applied this year.

Soil temperature and precipitation were the primary drivers of N₂O fluxes in agricultural systems, with warmer and wetter conditions leading to increased emissions. Both ADSH and USH significantly improved yield compared to the control, but their effects on N2O emissions were indirect, influenced more by environmental conditions than the treatments themselves. Contrary to the hypothesis, no significant difference in NUE was found between ADSH and USH, suggesting that long-term application of these fertilizers results in similar N availability. While ADSH has been hypothesized to reduce N₂O emissions due to its altered chemical composition and microbial accessibility, these results do not support this. This raises the question of whether the expected mechanisms—such as differences in nitrification and denitrification rates-are as influential under the cool, humid conditions of this study. It is possible that other factors, such as soil organic matter dynamics or microbial adaptation, play a stronger role in regulating emissions. Further research is needed to clarify whether the hypothesized benefits of anaerobic digestion in reducing N2O emissions hold under different environmental conditions or whether the treatments genuinely have no meaningful difference in this context.

These findings underline the importance of N management strategies tailored to Norway's climatic conditions, where temperature and precipitation fluctuations strongly influence N₂O emissions. Mitigation measures such as optimizing N application timing, improving drainage, adjusting soil pH, and adopting sitespecific tillage practices could help reduce agricultural N₂O emissions while maintaining soil health. Since ADSH and USH performed similarly in terms of yield, NUE, and N₂O emissions, the choice between these fertilizers may depend on other factors, such as access to anaerobic digesters, the demand for biogas, and broader sustainability goals in agriculture.

References

- Adviento-Borbe, M. A. A., Haddix, M. L., Binder, D. L., Walters, D. T., & Dobermann, A. (2007). Soil greenhouse gas fluxes and global warming potential in four high-yielding maize systems. Global Change Biology, 13(9), 1972–1988. <u>https://doi.org/10.1111/j.1365-2486.2007.01421.x</u>
- Auguie B (2017). _gridExtra: Miscellaneous Functions for "Grid" Graphics_. R package version 2.3, <u>https://CRAN.R-project.org/package=gridExtra</u>
- Barłóg, P., Hlisnikovský, L., & Kunzová, E. (2020). Effect of digestate on soil organic carbon and plant-available nutrient content compared to cattle slurry and mineral fertilization. Agronomy, 10(3), 379. https://doi.org/10.3390/agronomy10030379
- Bernal, M. P., Sommer, S. G., Chadwick, D., Qing, C., Guoxue, L., & Michel, F. C. (2017). Current approaches and future trends in compost quality criteria for agronomic, environmental, and human health benefits. In D. L. Sparks (Ed.), *Advances in Agronomy* (Vol. 144, pp. 143–233). Academic Press. <u>https://doi.org/10.1016/bs.agron.2017.03.002</u>
- Bleken, M. A., & Rittl, T. F. (2022). Soil pH-increase strongly mitigated N2O emissions following ploughing of grass and clover swards in autumn: A winter field study. Science of The Total Environment, 828, 154059. <u>https://doi.org/10.1016/j.scitotenv.2022.154059</u>
- Butterbach-Bahl, K., Baggs, E. M., Dannenmann, M., Kiese, R., & Zechmeister-Boltenstern, S. (2013). Nitrous oxide emissions from soils: How well do we understand the processes and their controls? *Philosophical Transactions of the Royal Society B: Biological Sciences*, 368(1621), 20130122. https://doi.org/10.1098/rstb.2013.0122
- Cao, Y., Wang, X., Bai, Z., Chadwick, D., Misselbrook, T., Sommer, S. G., Qin, W., & Ma, L. (2019). Mitigation of ammonia, nitrous oxide and methane emissions during solid waste composting with different additives: A metaanalysis. *Journal of Cleaner Production*, 235, 626– 635. <u>https://doi.org/10.1016/j.jclepro.2019.06.288</u>
- Clarke, E., Sherrill-Mix, S., & Dawson, C. (2023). ggbeeswarm: Categorical scatter (violin point) plots. R package version 0.7.2. <u>https://CRAN.R-project.org/package=ggbeeswarm</u>

- Dobermann, A., & Cassman, K. G. (2004). Environmental dimensions of fertilizer nitrogen: What can be done to increase nitrogen use efficiency and ensure global food security? In A. Mosier, J. K. Syers, & J. R. Freney (Eds.), Agriculture and the nitrogen cycle: Assessing the impacts of fertilizer use on food production and the environment (pp. 261–278). Washington, DC: Island Press.
- Douglas Bates, Martin Maechler, Ben Bolker, Steve Walker (2015). Fitting Linear Mixed-Effects Models Using Ime4. Journal of Statistical Software, 67(1), 1-48. <u>https://doi.org/10.18637/jss.v067.i01</u>
- El Mashad, H. M., Barzee, T. J., Franco, R. B., Zhang, R., Kaffka, S., & Mitloehner, F. (2023). Anaerobic Digestion and Alternative Manure Management Technologies for Methane Emissions Mitigation on Californian Dairies. Atmosphere, 14(1), 120. <u>https://doi.org/10.3390/atmos14010120</u>
- Forster, P., T. Storelvmo, K. Armour, W. Collins, J.-L. Dufresne, D. Frame, D.J. Lunt, T. Mauritsen, M.D. Palmer, M. Watanabe, M. Wild, and H. Zhang, 2021: The Earth's Energy Budget, Climate Feedbacks, and Climate Sensitivity. In Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change [Masson-Delmotte, V., P. Zhai, A. Pirani, S.L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M.I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J.B.R. Matthews, T.K. Maycock, T. Waterfield, O. Yelekçi, R. Yu, and B. Zhou (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, pp. 923–1054, https://doi.org/10.1017/9781009157896.009
- Frick, H., Efosa, N., Oberson, A., Krause, H.-M., Nägele, H.-J., Frossard, E., & Bünemann, E. K. (2024). Nitrogen dynamics after slurry application as affected by anaerobic digestion, biochar and a nitrification inhibitor. Soil Use and Management, 40, e12953. <u>https://doi.org/10.1111/sum.12953</u>
- Fuss R (2024). _gasfluxes: Greenhouse Gas Flux Calculation from Chamber Measurements_. R package version 0.7, <u>https://CRAN.R-project.org/package=gasfluxes</u>
- García-López, A. M., Delgado, A., Anjos, O., & Horta, C. (2023). Digestate Not Only Affects Nutrient Availability but Also Soil Quality Indicators. Agronomy, 13(5), 1308. <u>https://doi.org/10.3390/agronomy13051308</u>
- 16. Garrett Grolemund, Hadley Wickham (2011). Dates and Times Made Easy with lubridate. Journal of Statistical Software, 40(3), 1-25. URL https://www.jstatsoft.org/v40/i03/
- 17. Kassambara, A. (2023). ggpubr: 'ggplot2' based publication ready plots. R package version 0.6.0. https://CRAN.R-project.org/package=ggpubr
- Kuznetsova, A., Brockhoff, P. B., & Christensen, R. H. B. (2017). *lmerTest* package: Tests in linear mixed effects models. *Journal of Statistical Software*, 82(13), 1–26. <u>https://doi.org/10.18637/jss.v082.i13</u>
- Ladha, J. K., Pathak, H., Krupnik, T. J., Six, J., & van Kessel, C. (2005). Efficiency of fertilizer nitrogen in cereal production: Retrospects and prospects. Advances in Agronomy, 87, 85-156. <u>https://doi.org/10.1016/S0065-2113(05)87003-8</u>

- 20. Laiq Ur Rehman, M., Iqbal, A., Chang, C.-C., Li, W., & Ju, M. (2019). Anaerobic digestion. Water Environment Research, 91(10), 1253–1271. <u>https://doi.org/10.1002/wer.1219</u>
- Laurent Philippot, Sara Hallin, Michael Schloter, Ecology of Denitrifying Prokaryotes in Agricultural Soil, Advances in Agronomy, Academic Press, Volume 96, 2007, Pages 249-305, ISBN9780123742063, https://doi.org/10.1016/S0065-2113(07)96003-4
- 22. Lenth, R. (2024). *emmeans: Estimated marginal means, aka least-squares means*. R package version 1.10.5. <u>https://CRAN.R-project.org/package=emmeans</u>
- Li L, Zheng Z, Wang W, et al. Terrestrial N2O emissions and related functional genes under climate change: A global meta-analysis. Glob Change Biol. 2020; 26: 931–943. <u>https://doi.org/10.1111/gcb.14847</u>
- 24. Lüdecke et al., (2021). performance: An R Package for Assessment, Comparison and Testing of Statistical Models. Journal of Open Source Software, 6(60), 3139. <u>https://doi.org/10.21105/joss.03139</u>
- Møller, H.B., Sørensen, P., Olesen, J.E., Petersen, S.O., Nyord, T., & Sommer, S.G. (2022). Agricultural Biogas Production—Climate and Environmental Impacts. *Sustainability*.
- Möller, K., & Müller, T. (2012). Effects of anaerobic digestion on digestate nutrient availability and crop growth: A review. Agronomy for Sustainable Development, 32(1), 225-238. https://doi.org/10.1007/s13593-011-0038-8
- Ngatia, L., Grace III, J. M., Moriasi, D., & Taylor, R. (2019). Nitrogen and phosphorus eutrophication in marine ecosystems. In H. B. Fouzia (Ed.), Monitoring of marine pollution (pp. 77–95). IntechOpen. <u>https://doi.org/10.5772/intechopen.81869</u>
- NIBIO. (n.d.). Meteorological data for Tingvoll (January 2015– January 2025). Norwegian Institute of Bioeconomy Research (NIBIO). Retrieved February 5th 2025 from <u>https://lmt.nibio.no/stationinfo/49/</u>
- Niskanen, M., Niemeläinen, O., Jauhiainen, L., & Kurri, O. (2006). Effect of quality assessment on cultivar performance in timothy variety testing in Finland. In T. Sveinsson (Ed.), Timothy productivity and forage quality: Possibilities and limitations (pp. 107–110). Proceedings of NJF Seminar 384, Akureyri, Iceland
- 30. Nyborg, S. B. 1995. Biologisk nitrogenfiksering. Småskrift 6/95. Infosenteret, Forskningsparken i Ås AS
- Omara, P., Aula, L., Oyebiyi, F. and Raun, W.R. (2019), World Cereal Nitrogen Use Efficiency Trends: Review and Current Knowledge. Agrosystems, Geosciences & Environment, 2: 1-8 180045. https://doi.org/10.2134/age2018.10.0045
- Qiu, Q., Mgelwa, A. S., Jin, S., & Hu, Y. (2022). Nitrogen-induced changes in soil environmental factors are more important than nitrification and denitrification gene abundance in regulating N₂O emissions in subtropical forest soils. *Frontiers in Plant Science*, 13, 950367. <u>https://doi.org/10.3389/fpls.2022.950367</u>
- 33. R Core Team (2024). _R: A Language and Environment for Statistical Computing_. R Foundation for Statistical Computing, Vienna, Austria. <u>https://www.R-project.org/</u>

- Rapson, T., & Dacres, H. (2013). Analytical techniques for measuring nitrous oxide. *Trends in Analytical Chemistry*, 54,65-74. <u>https://doi.org/10.1016/j.trac.2013.11.004</u>
- 35. Rittl, T. F., de Boer, A., Pommeresche, R., & Løes, A.-K. (2024). Data of slurry quality, soil chemistry, and crop yield and composition from a field experiment established 2011 comparing digested and undigested slurry. Data in Brief, 53, 110201. <u>https://doi.org/10.1016/j.dib.2024.110201</u>
- Rittl, T., Pommeresche, R., Johansen, A. *et al.* Anaerobic digestion of dairy cattle slurry—long-term effects on crop yields and chemical soil characteristics. *Org. Agr.* 13, 547–563 (2023). <u>https://doi.org/10.1007/s13165-023-00447-0</u>
- Six, J., Ogle, S.M., Jay breidt, F., Conant, R.T., Mosier, A.R. and Paustian, K. (2004), The potential to mitigate global warming with no-tillage management is only realized when practised in the long term. Global Change Biology, 10: 155-160. <u>https://doi.org/10.1111/j.1529-8817.2003.00730.x</u>
- 38. Smith, K.A., Ball, T., Conen, F., Dobbie, K.E., Massheder, J. and Rey, A. (2018), Exchange of greenhouse gases between soil and atmosphere: interactions of soil physical factors and biological processes. Eur J Soil Sci, 69: 10-20. <u>https://doi.org/10.1111/ejss.12539</u>
- 39. Tambone, F., Scaglia, B., D'Imporzano, G., Schievano, A., Orzi, V., Salati, S., & Adani, F. (2010). Assessing amendment and fertilizing properties of digestates from anaerobic digestion through a comparative study with digested sludge and compost. Chemosphere, 81(5), 577–583. https://doi.org/10.1016/j.chemosphere.2010.08.034
- Tian, H., Pan, N., Thompson, R. L., Canadell, J. G., Suntharalingam, P., Regnier, P., Davidson, E. A., Prather, M., Ciais, P., Muntean, M., Pan, S., Winiwarter, W., Zaehle, S., Zhou, F., Jackson, R. B., Bange, H. W., Berthet, S., Bian, Z., Bianchi, D., Bouwman, A. F., Buitenhuis, E. T., Dutton, G., Hu, M., Ito, A., Jain, A. K., Jeltsch-Thömmes, A., Joos, F., Kou-Giesbrecht, S., Krummel, P. B., Lan, X., Landolfi, A., Lauerwald, R., Li, Y., Lu, C., Maavara, T., Manizza, M., Millet, D. B., Mühle, J., Patra, P. K., Peters, G. P., Qin, X., Raymond, P., Resplandy, L., Rosentreter, J. A., Shi, H., Sun, Q., Tonina, D., Tubiello, F. N., van der Werf, G. R., Vuichard, N., Wang, J., Wells, K. C., Western, L. M., Wilson, C., Yang, J., Yao, Y., You, Y., and Zhu, Q. (2024). Global nitrous oxide budget (1980–2020), Earth Syst. Sci. Data, 16, 2543–2604, <u>https://doi.org/10.5194/essd-16-2543-2024</u>
- 41. TL, Miller E, Bache SM, Müller K, Ooms J, Robinson D, Seidel DP, Spinu V, Takahashi K, Vaughan D, Wilke C, Woo K, Yutani H (2019). "Welcome to the tidyverse." _Journal of Open Source Software_, *4*(43), 1686. doi:10.21105/joss.01686 https://doi.org/10.21105/joss.01686
- 42. Ulrich Halekoh, Søren Højsgaard (2014). A Kenward-Roger Approximation and Parametric Bootstrap Methods for Tests in Linear Mixed Models - The R Package pbkrtest. Journal of Statistical Software, 59(9), 1-30. <u>https://www.jstatsoft.org/v59/i09/</u>
- 43. Walsh, J. J., Jones, D. L., Chadwick, D., & Williams, A. (2018). Repeated application of anaerobic digestate, undigested cattle slurry and inorganic

fertilizer N: Impacts on pasture yield and quality. *Grass and Forage Science*, 73, 758-763. <u>https://doi.org/10.1111/gfs.12354</u>

- Wei, Z., Ying, H., Guo, X., Zhuang, M., Cui, Z., & Zhang, F. (2020). Substitution of Mineral Fertilizer with Organic Fertilizer in Maize Systems: A Meta-Analysis of Reduced Nitrogen and Carbon Emissions. Agronomy, 10(8), 1149. <u>https://doi.org/10.3390/agronomy10081149</u>
- 45. Wickham H, Averick M, Bryan J, Chang W, McGowan LD, François R, Grolemund G, Hayes A, Henry L, Hester J, Kuhn M, Pedersen
- 46. Wickham, H., & Bryan, J. (2023). *readxl: Read Excel files*. R package version 1.4.3. <u>https://CRAN.R-project.org/package=readxl</u>
- 47. World Bank. (n.d.). *Climate change knowledge portal: Norway*. The World Bank. Retrieved October 22nd 2024, from <u>https://climateknowledgeportal.worldbank.org/country/norway</u>
- Yang, L., Deng, Y., Wang, X., Zhang, W., Shi, X., Chen, X., Lakshmanan, P., & Zhang, F. (2021). Global direct nitrous oxide emissions from the bioenergy crop sugarcane (Saccharum spp. inter-specific hybrids). Science of The Total Environment, 752, 141795. <u>https://doi.org/10.1016/j.scitotenv.2020.141795</u>

Popular science summary

Agriculture is a major source of greenhouse gas emissions, particularly nitrous oxide (N₂O), which has a much stronger warming effect than carbon dioxide. As the world looks for ways to reduce its environmental impact, finding more sustainable farming practices is crucial. Fertilizer management plays a key role in this, as inefficient use of fertilizers can both harm the environment and limit crop growth. In Norway, where organic farming is gaining popularity, understanding how different types of fertilizers affect both crop yield and emissions is essential for sustainable agricultural practices. This study examined the effects of two types of cattle slurry—anaerobically digested slurry (ADSH) and untreated slurry (USH)—on crop yield, nitrogen use efficiency (NUE), and N₂O emissions in organic dairy farming in Norway.

The results showed that both ADSH and USH improved crop yield compared to the control, with no significant difference between the two. N₂O emissions were primarily driven by soil temperature and rainfall, rather than fertilizer type. While the long-term application of ADSH did not reduce N₂O emissions as expected, the findings suggest that environmental factors like temperature and precipitation have a larger influence on emissions than the fertilizer type itself. Additionally, the presence of clover in the field reduced nitrogen efficiency, as clover fixes nitrogen naturally, affecting NUE.

These findings highlight the importance of considering environmental conditions in nitrogen management strategies. Both ADSH and USH can be effective fertilizers, with the choice between them depending on other factors such as access to biogas production or broader sustainability goals in agriculture.

Acknowledgements

I would like to express my sincere gratitude to my supervisors, Sofia Delin from SLU and Tatiana Rittl from NORSØK, for their invaluable guidance and support throughout this project. A special thank you to Johanna Zimmermann from Rheinland-Pfälzische Technische Universität Kaiserslautern-Landau, whose assistance with data analysis and writing was essential to this work. I am also deeply grateful to my family for their support: my stepfather, Luc Bussière, my mother, Ane Timenes Laugen, and my father, Geir Løe, for their help with statistics and writing. Finally, I would like to thank NORSØK (Norwegian Centre for Organic Agriculture) for providing the opportunity to collaborate on this project and for their ongoing support throughout my research.

Publishing and archiving

Approved students' theses at SLU are published electronically. As a student, you have the copyright to your own work and need to approve the electronic publishing. If you check the box for **YES**, the full text (pdf file) and metadata will be visible and searchable online. If you check the box for **NO**, only the metadata and the abstract will be visible and searchable online. Nevertheless, when the document is uploaded it will still be archived as a digital file. If you are more than one author, the checked box will be applied to all authors. You will find a link to SLU's publishing agreement here:

• <u>https://libanswers.slu.se/en/faq/228318</u>.

 \boxtimes YES, I/we hereby give permission to publish the present thesis in accordance with the SLU agreement regarding the transfer of the right to publish a work.

 \Box NO, I/we do not give permission to publish the present work. The work will still be archived and its metadata and abstract will be visible and searchable.